Author's response to reviews

Title: Primary hydatid cyst of the adrenal gland resected laparoscopically: a case report.

Authors:

Gianlorenzo Dionigi (gianlorenzo.dionigi@uninsubria.it)
Gianpaolo Carrafiello (gcarraf@tin.it)
Chiara Recaldini (chiarec@libero.it)
Fausto Sessa (fausto.sessa@uninsubria.it)
Luigi Boni (luigi.boni@uninsubria.it)
Francesca Rovera (francesca.rovera@uninsubria.it)
Renzo Dionigi (renzo.dionigi@uninsubria.it)

Version: 7 Date: 1 August 2007

Author's response to reviews:

Dear Editor

following are the details of the revision:

Title - Is now identical in both the manuscript and submission system. No the full stop

Author list - The list of authors in the manuscript are written exactly as they are in the submission system. We remove the authors' qualifications/titles from the manuscript and submissions system.

Title page as well as Affiliations were modified according to You

Abstract is identical in both the manuscript and submission system.

We remove the keywords

We change: Competing interest -> Competing interests
Acknowledgement -> Acknowledgements

Figures - figure 1b was submitted as figure 2 and figure 2 as figure 3 etc.

Thank You very much
Gianlorenzo Dionigi